Author's response to reviews

Title: Mandatory chromosomal segment balance in aneuploid tumor cells

Authors:

Maria Kost-Alimova (Maria_Kost-Alimova@mtc.ki.se)
Eva Darai-Ramqvist (Eva_Darai@mtc.ki.se)
Wing L Yau (boywl@ust.hk)
Agneta Sandlund (Agneta.Sandlund@mtc.ki.se)
Ludmila Fedorova (rosesinaz@msn.com)
Ying Yang (yyang1026@yahoo.com)
Irina Kholodnyuk (irina.kholodnyuk@ebc.uu.se)
Yue Cheng (chengy@mail.nih.gov)
Maria L Lung (bomaria@ust.hk)
Eric Stanbridge (ejstanbr@uci.edu)
George Klein (Georg.Klein@mtc.ki.se)
Stefan Imreh (Stefan.Imreh@mtc.ki.se)

Version: 2 Date: 13 November 2006

Author's response to reviews:

I have made changes, which allow transfer from BMC Genomics to BMC Cancer. The changes include:
1. Addition of "Methods" part into Abstract;
2. Addition of "Competing interests" section into the text

I resubmit our manuscript following the instructions provided.

Sincerely yours,

Maria Kost-Alimova